Author’s response to reviews

Title: Assessment of two complementary influenza surveillance systems: Sentinel primary care Influenza-like Illness versus severe hospitalized laboratory-confirmed influenza using the moving epidemic method

Authors:

Nuria Torner (nuria.torner@gencat.cat)
Luca Basile (chags@gencat.cat)
Ana Martinez (a.martinez@gencat.cat)
Cristina Rius (crius@aspb.cat)
Pere Godoy (pere.godoy@gencat.cat)
Mireia Jane (mireia.jane@gencat.cat)
Angela Dominguez (angela.dominguez@ub.edu)

Version: 1 Date: 07 Nov 2018

Author’s response to reviews:

In accordance with suggested changes as to formatting guidelines, the Declarations section and the mandatory sub-sections as listed below have been included:

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors’ Contributions
- Acknowledgements
- Authors’ Information

A new version of the previous with rephrasing of some parts of previous manuscript is submitted